Author's response to reviews

Title: Cross-checking to reduce adverse events resulting from medical errors in the emergency department: rationale and methodology of the CHARMED cluster randomised study.

Authors:

Yonathan Freund (yonatman@gmail.com)
Alexandra Rousseau (alexandra.rousseau@sat.aphp.fr)
Laurence Berard (laurence.berard@sat.aphp.fr)
Helene Goulet (helen.goulet@gmail.com)
Patrick Ray (patrick.ray@tnn.aphp.fr)
Benjamin Bloom (ben.bloom@nhs.net)
Tabassome Simon (tabassome.simon@sat.aphp.fr)
Bruno Riou (bruno.riou@psl.aphp.fr)

Version: 2 Date: 15 June 2015

Author's response to reviews: see over
Dear editor,

Please find attached a revised version of our manuscript « rationale and methodology of the CHARMED study ».

There might have been a misunderstanding: as this is only the method paper, we cannot provide all that was required after your initial assessment. We hope that you will find this version more suitable.

Yours faithfully

1. CONSORT guidelines:

We added the CONSORT checklist as appendix, however, as we do not report results of the trial, CONSORT cannot be entirely filled.

2. Competing Interests:

We added this paragraph

3. Authors’ Contributions:

We added this paragraph

4. Acknowledgement:

We have no acknowledgement to mention

5. List of Abbreviations:

We added this paragraph